Author's response to reviews

Title: Unilateral Primitive Hydatid Cyst of Scrotal Case report

Authors:

Ahmed Amine Bouchikhi (drbouchikhi@gmail.com)
Driess Amiroune (d.amiroune@gmail.com)
MF TAZI (mfadltazi@gmail.com)
S Mellass (drsmellas@yahoo.fr)
JE Elammari (jeelammari@yahoo.fr)
MJ Elfassi (mjelfassi@hotmail.fr)
A Khallouk (drakhllouk@yahoo.fr)
MH Farih (mhfarih@hotmail.fr)
y alaoui (abouchikhi564@gmail.com)

Version: 3 Date: 18 December 2012

Author's response to reviews: see over
Dear Editor of JMCR,

We are submitting our paper *Unilateral Scrotal Primitive Hydatid Cyst case report*

We are estimating that this paper would contribute on clarifying attitudes in the elucidated disease.

We do hope that our manuscript will find the way to publication in your journal and kind regards,

Dr. Ahmed Amine Bouchikhi